

Open Archive Toulouse Archive Ouverte (OATAO)

OATAO is an open access repository that collects the work of some Toulouse researchers and makes it freely available over the web where possible.

This is a publisher's version published in: https://oatao.univ-toulouse.fr/27339

Official URL:https://doi.org/10.3390/s20164532

To cite this version :

Feriol, Florent and Vivet, Damien and Watanabe, Yoko A Review of Environmental Context Detection for Navigation Based on Multiple Sensors. (2021) Sensors, 20 (16). 4532. ISSN 1424-8220

Any correspondence concerning this service should be sent to the repository administrator: <u>tech-oatao@listes-diff.inp-toulouse.fr</u>



Review



A Review of Environmental Context Detection for Navigation Based on Multiple Sensors

Florent Feriol ^{1,*}, Damien Vivet ¹, and Yoko Watanabe ²

- ¹ Optronics and Signal Research Group, ISAE-SUPAERO, 31055 Toulouse, France; damien.vivet@isae-supaero.fr
- ² Department of Information Processing and Systems, ONERA, 31055 Toulouse, France; yoko.watanabe@onera.fr
- * Correspondence: florent.feriol@isae-supaero.fr

Received: 9 June 2020; Accepted: 6 August 2020; Published: 13 August 2020



Abstract: Current navigation systems use multi-sensor data to improve the localization accuracy, but often without certitude on the quality of those measurements in certain situations. The context detection will enable us to build an adaptive navigation system to improve the precision and the robustness of its localization solution by anticipating possible degradation in sensor signal quality (GNSS in urban canyons for instance or camera-based navigation in a non-textured environment). That is why context detection is considered the future of navigation systems. Thus, it is important firstly to define this concept of context for navigation and to find a way to extract it from available information. This paper overviews existing GNSS and on-board vision-based solutions of environmental context detection. This review shows that most of the state-of-the art research works focus on only one type of data. It confirms that the main perspective of this problem is to combine different indicators from multiple sensors.

Keywords: context detection; environmental context detection; context-aware navigation; UGV; unmanned ground vehicle; vision; image processing; GNSS signal; sky extraction; scene analysis; aerial photography segmentation; remote sensing; signal processing

1. Introduction

In the past decade, much work has been done to make navigation more precise and more reliable: sensor fusion, improvement of signal quality, receiver hardware enhancement (antennas arrays for instance), mitigation of errors, robust filtering, and so on. Fusing data from different sensors is a good way to make a navigation system more accurate, but it could also degrade the navigation accuracy when some of those data are erroneous. For instance, fusing Global Navigation Satellite System (GNSS) data with vision-based data in an indoor situation or in a harsh environment would create positioning errors since GNSS data might be unreliable in such a context. That is why context information is interesting and useful for navigation: behavioural and environmental contexts. The behavioural context represents the activity of either a vehicle in the case of robotic applications such as Unmanned Ground/Aerial Vehicles (UGVs or UAVs) (for example, accelerating/slowing down) or a user in the case of a smartphone (climbing stairs, running, taking a lift, etc.). The environmental context defines the surroundings, like indoor/outdoor, close to a building, under a tree, and so on. Reference [1] is an interesting introduction to environmental contexts since it gives an exhausting list including even the space or submarine environment.

Context-aware navigation is a hot topic for pedestrian applications where human behaviour is detected and classified mainly using Inertial Measurement Unit (IMU) sensor data available in

smartphones [3–5] (see also Section 4). However, when limiting our scope to on-/near-ground vehicle navigation, the behavioural context is of less interest (even if, in recent years, some papers have shown that it is possible to use the behavioural context to help environmental context detection [3,6,7]). Instead, as illustrated in the previous example of GNSS, the environmental context can give important information about the potential quality of certain navigation sensor signals. In the literature today, environmental context awareness is mostly used in GNSS applications for satellite mitigation, but not in a more general framework of navigation. In addition, few works focus on multi-sensor fusion for context detection. Therefore, the objective of this paper is to explore the multi-sensor solutions to robustly classify the environmental contexts that have an impact on the sensor quality. Such context detection could be used to adapt data fusion in cases where one (or more) of the sensor data is possibly degraded (e.g., GNSS in indoor situations, a camera in a non-textured environment). It could also help to choose a proper data processing algorithm considering the expected sensor signal or data quality in a particular context. This paper assumes a classical onboard sensor set for ground vehicle (or drone) navigation: GNSS receiver, IMU, and camera(s). Indeed, adding extra sensors would augment the context detection capability, but on the other hand, it is very important to maintain the viability of the system. For this reason, this paper primarily focuses on a review of environmental context detection methods based on these most frequently used navigation sensors. Nonetheless, we will give a glance at methods based on other types of sensors in Section 4. Figure 1 illustrates a scheme of the context-aware adaptive navigation system with the considered sensor set, and Table 1 lists the environmental contexts along with their impacts on the GNSS and vision sensors' quality. As shown in Figure 1, adaptation could be applied both to data extraction in each sensor system and to data fusion in the navigation filter (a list of different solutions is available in Table 1).



Figure 1. Context-aware adaptive navigation concept.

	GNSS	5		Articles	
Context	Impact	Adaptation	Impact	Adaptation	
Urban canyon (narrow street with tall buildings) Signal not available/ high positioning errors		Tight-coupling, NLOS filtering, shadow matching	None	Point- or line-based feature extraction	[8–13]
Dense urban area (residential area)	High NLOS and multipath risk	Tight-coupling, NLOS filtering	None	Classical point- or line-based feature extraction	[10-18]
Low density urban area (suburban area)	Low NLOS risk, but multipath effect possible	Low NLOS risk, Doppler aiding, but multipath multipath mitigation, None effect possible loose-coupling		Classical point- or line-based feature extraction	[10–18]
Deep indoor (no line of sight to the exterior)	Signal not available	available Vision/INS coupling few robust point features Line-ba		Line-based feature extraction	[1,9,16,19–33]
Light indoor (close to door, window, or balcony, also called semi-indoor)	Signal with high errors (due to both attenuation and reflection)	Vision/INS coupling	Lake of texture, few robust point features, glare effect	Line-based feature extraction, additional image processing step	[1,9,16,19–33]
Open sky	Open sky Perfect quality of the signal GNSS/INS loose-coupling		None	Switch off	[1,8,10–13,15,34]
Dense forest	Signal attenuated and multipath	Extension of coherent integration time, Doppler aiding, multipath mitigation	Unstructured Combination of points environment and colour features		[15–18]
Light forest (couple of trees)	orestExtension of coherentof trees)Signal attenuatedintegration time,Doppler aiding		Unstructured environment, glare effect	Combination of points and colour features, additional image processing step	[15–18]
Near water surface	Near water surface Tremendous number of reflections Doppler aiding, multipath mitigation		No texture and landmarks	Switch off	[16]

Table 1. List of the environmental contexts of interest, their impacts on GNSS/vision sensors, and the corresponding articles.

Indeed, some algorithms for data extraction are designed to be more successful under particular conditions. For instance, algorithms used to extract navigation information from camera images can be selected depending on the context. In indoor or urban environments, which mainly contain structured buildings and handmade objects, algorithms based on line detection as a key feature could be more appropriate and efficient. In outdoor environments such as natural areas, algorithms should rather be based on point feature detection, as explained in [35].

The quality of the GNSS signals also significantly depends on the environment configuration. The GNSS signals can be attenuated in urban environments, but also under trees, as shown in [36–38]. In such conditions where the signal is weak, the coherent integration time duration can be extended to improve the signal detection [39]. Another known problem of GNSS in harsh environment is the potential signal reflections on surfaces. This changes the travel distance from the satellite to the receiver. This effect creates an error in the position solution calculated by the receiver. Two different scenarios exist: multipaths and None Line Of Sight (NLOS). Examples of multipath and NLOS situations are shown in Figure 2. In NLOS scenarios, one (or more) GNSS signal is masked in the direct path, but reaches the receiver via a reflected path. Most of the time, the signal blocking element is a tall structure (e.g., a building), which means that NLOS exists primarily in urban areas. Since NLOS creates higher errors of positioning than the multipath effect [40], it is important to find a way to mitigate this problem. One solution is to modify and adapt the way GNSS data are fused with the other sensors in the navigation filter. The tight-coupling, which uses the pseudoranges and the Doppler estimations, has the advantage of needing fewer available satellites (down to two), which is very beneficial in cluttered environments such as urban canyons where many satellites can be NLOS. On the contrary, the loose-coupling is simpler and thus is more efficient in terms of computational load, but performs poorer in harsh environments where the position cannot be easily computed with pseudoranges. It is also more robust to GNSS and INS faults since one cannot impact the other. Another solution is the NLOS satellites' filtering thanks to image processing, as explained in Section 3.1, which can also be useful to limit the near-far effect [34]. The multipath effect scenario happens when there is a combination of the direct and reflected path signals (Figure 2). Reference [41] explained that there are three different types of multipath effects:

- Diffuse multipath typically happens when the signal encounters a cluttered metallic surface such as overhead wires. The signal is diffused in a wide variety of directions, creating an error of positioning, which can go up to 10 m.
- Specular multipath appears with reflective surfaces such as a mirror and glass and can lead to positioning errors between two and six m.
- Water reflections are linked to the presence of a water surface next to the antenna and can create positioning errors of an order of 10 m.



Figure 2. Schemes of the multipath and NLOS effect.

Multipath mitigation can be done thanks to maximum-likelihood methods such as MMT [42] (Multipath Mitigation Technology), MEDLL [43] (Multipath Estimation Delay Lock Loop), and FIMLA [44] (Fast Iterative Maximum Likelihood Algorithm), but require much computational power. It is also possible to use Doppler measurements as additional information since multipath has a lower impact on them than on pseudoranges [45,46]. Doppler methods can also be used in situations where the GNSS signal is very noisy (under a canopy, semi-indoor areas, etc.). Other GNSS limitations that are not really linked to a particular environmental context, such as jamming/spoofing, will not be considered in this paper.

If there is a means to detect the environment context correctly, those aforementioned adaptation schemes can be applied to the navigation system. Such context-aware adaptive navigation solutions will improve the navigation performance in terms of both the accuracy and the reliability, which can help current navigation applications or even lead to new ones. As presented later in this review, most of the existing context-aware solutions base their context detection only on one type of indicator from one sensor (GNSS or vision) and are basically designed to do sensor management in the binary indoor/outdoor situation. Unlike these, our research interest and perspective is to propose new multi-indicator/multi-sensor-based solutions that are capable of detecting and classifying the environmental contexts more finely and robustly for the navigation adaptation purpose.

The paper is organized as follows: Section 2 presents different context indicators derived from the GNSS signals. Section 3 focuses on context detection by vision sensors. Section 4 briefly introduces techniques that are based on others sensors. Finally, Section 5 provides the summary and our perspectives towards context detection for navigation.

2. GNSS Signal-Based Context Indicators

It is natural to think of using the characteristics of the GNSS signal itself to detect the context affecting it. Different indicators can be used for this purpose and are presented in the following parts.

2.1. C/N0

The Carrier to Noise ratio (C/N0) is probably the most well-known indicator of the GNSS signal. It is a signal-to-noise ratio that is not quantified in dB, but in dB.Hz. It gives information on the signal quality and on how badly (or not) the signal has been attenuated. According to [34], a standard C/N0 is around 45 dB.Hz in an open sky area. They also showed that different ranges of C/N0 values can be representative of a specific context (Table 2).

	Outdoor	Soft-Indoor	Intermediate	Deep-Indoor
C/N0 (dB.Hz)	35–45	25–35	10–25	<10

Table 2. Carrier to Noise ratio (C/N0) ranges for different contexts (extracted from [34]).

When looking at this table, it seems very easy to distinguish indoor environments from outdoor ones: a simple threshold seems more than enough. However, the reality is much more complex. Firstly, C/N0 is not that stable and reliable. This was showcased in [1], where they acquired 100 s of C/N0 values (at a frequency of 1Hz) in different contexts such as indoors, outdoors in urban/residential areas, and open field. For each of those recordings, the authors computed histograms of C/N0. Those histograms showed that C/N0 values vary greatly during the period of acquisition (due to temporal losses of satellite visibility for a short time period). As a conclusion, we can fairly say that it is not possible to implement a simple threshold, as given in Table 2, to detect a context. However, the authors of [1] made two observations based on those data:

- The average C/N0 value is higher in outdoor environments than in indoor ones.
- The standard deviation is larger in outdoor environments than in indoor ones, which makes sense
 since signal occlusion is more likely to happen in outdoor environments because a small change
 in the satellite constellation can greatly affect the satellite visibility in a cluttered environment.

Those indicators of C/N0 (average and variance) seem useful to make a classification of the indoor/outdoor context. The problem is that we need an integration time long enough to obtain reliable values for these indicators. When considering the vehicle navigation application, the GNSS receiver is not static and travels in the environment during the data sampling time window. This displacement can create the unreliability of C/N0 values. Secondly, the definitions of the soft-indoor and intermediate context are not very clear. A GNSS receiver can be for instance placed at the exit of a building and therefore should be considered as intermediate, but the same classification can happen with a receiver close to a window, which should be classified as soft-indoor. This means that GNSS indicators cannot classify context robustly in certain situations where the performances are variable.

The problem of movement was addressed in [19] where they built a setup with the receiver antenna on the back of a human who was moving between indoor and outdoor locations. In the recorded data, there was a clear difference of the C/N0 values when passing from outdoors to indoors. There was a diminution of the average value, as well as an increase in noise (higher variance). This noise makes it difficult to detect a transition between outdoors and indoors with a simple threshold. The authors of [19] proposed a solution based on a Rice law (probability distribution), which is described by a factor K. The goal is to estimate the value of K, which corresponds to the fading of the signal and is in dB. More details on how to estimate the K-Rician factor can be found in [47]. Thanks to the *K*-Rician factor, most of the noise is removed and the signal becomes much more stable. It is then possible to classify rather easily the indoor and outdoor contexts. To achieve this classification, two thresholds were used in the paper. If C/N0 < 32 dB-Hz and K < 10 dB, then the context was declared as indoor. The authors also tried different sampling times: 100 ms, 500 ms, and 1 s. At the end, the sampling time of one second gave the best result, which was not very surprising considering the amount of noise on the 100 ms signal. In [48], they also showed that the Rician-factor method works for various velocities (obviously, the higher the velocity is, the higher the variations will be). However, this indoor/outdoor classification based on the K-Rician factor and C/N0 values was only used as an indicator to switch between processing strategies (vector-based and block-based), and the article did not mention the accuracy of the context classification, but only of the navigation solution as a whole.

Indoor/outdoor classification by the C/N0 indicator seems to perform well thanks to the *K*-Rician factor, but the article [20] went even beyond that and performed the context classification into indoors, outdoors, and urban. They used a bottom-up strategy where they firstly tried to classify indoors, hybrid, and outdoors. To do so, they implemented an HMM (Hidden Markov Model) with two features:

- The number of satellites with a C/N0 > 25 dB-Hz. This idea of the number of visible satellites was also exploited in [21] (SatProbe) to classify indoors from outdoors (only based on the GPS constellation).
- The sum of the C/N0 of satellites > 25 dB-Hz

Once the first classification was done and if the detected context was outdoors, a fuzzy inference system with the following two features was used to find whether the receiver was in an open sky or in an urban environment:

• The sum of squares of the pseudo-range residuals, which is defined as follows:

$$zPRR = \left(\sum_{i=1}^{N} |\rho_c - \rho_e|^2\right) / (N - 4)$$
(1)

with *N* the number of satellites, ρ_c the measured pseudo-range, and ρ_e the estimated range.

The sum of the C/N0 values of satellites >25 dB-Hz

The overall system had a success rate of 88.2%, which was a promising result. The limitation of this method is that the most challenging context in terms of classification was only described as intermediate. This means that we do not really know the nature of those intermediate contexts, and it leaves us with a big problematic: Can we still use GNSS reliably in those contexts? The authors also only used a unique constellation to build their indicators, and a multi-constellation solution could enhance the results. Another problem, which was not addressed in the previous papers, is the robustness of those methods since GNSS signal quality depends on the location and type receiver, its displacement velocity, atmospheric conditions/weather, and even time (since it has an influence on the constellation). As a consequence, the threshold values provided cannot be trusted in all scenarios.

2.2. Pseudo-Range

Another indicator that is used to classify LOS signals from NLOS signals is the pseudo-range residuals, as proposed in [49]. As defined in Section 2.1, the pseudo-range residuals are the difference between the measured pseudo-range (thanks to the satellite ephemeris data and the signal travelling time) and the estimated pseudo-range that is obtained once the user position has been computed. Being able to distinguish LOS from NLOS is interesting for the context detection since a large number of NLOS satellites means that the context is indoors or at least dense urban. The principle is simple: if a satellite is in an NLOS situation, its measured pseudo-range should be longer than the estimated one due to a longer signal path. One problem noticed in [49] is that pseudo-range residuals are biased due to the fact that the estimated user position is calculated from the measured pseudo-ranges. At least four satellites are needed in order to obtain a position solution, and some of them can be NLOS. Thus, the position admits some errors. To reduce this bias effect, the authors of [49] proposed to use the second derivative of the pseudo-range residuals. Variations of this indicator are much larger in the case of NLOS than in LOS and hence could help us to define if we are in an urban environment. However, the main issue is that we need a certain integration time to be sure to observe some important variations. Although this indicator cannot work alone, it can be used in some uncertain detection cases where additional data are needed to confirm the context. It is also noteworthy that the authors tried to use the Doppler shift to classify LOS from NLOS. Unfortunately, this solution does not really work in low or null velocity situations since the Doppler shift represents the relative speed between the receiver and the satellite.

2.3. Satellite Elevations

Satellite elevation is not an indicator that gives directly any information on the context. However, it can be used as supplementary information to help the context detection, like the pseudo-range. Indeed, the lower the satellite is, the higher the risk is to have its signal occluded by a building. Elevation can be used for detecting multipaths or for NLOS/LOS classification. To do so, it is mandatory to have a height map of the environment to compare buildings height with satellites elevations (azimuths are also used in this process). Formerly, such height maps needed to be created by various methods [50], but currently, 3D city models are widely available [51–54]. All those techniques do work, they require a 3D environment model and an accurate estimation of the receiver position, which are far from being available everywhere. It was also stated in [55] that those techniques require much computational resource, which cannot be supported by every navigation setup.

Being able to distinguish which signal can be blocked from a 3D map is useful for shadow matching. This technique improves the position precision solution by verifying which signals are receivable in the direct path to reduce the search space of the user position. It is useful especially in very cluttered areas such as urban canyons [56–58]. Indeed, if a direct satellite signal is supposed to be LOS, whereas the signal quality coming from that satellite is poor (sometimes, there is no signal at all), this means that the position solution calculated is incorrect and, thus, needs to be recalculated. A new candidate user position close to the initial one is computed by using areas where the satellite

is supposed to be blocked. This process can be refined by using other satellites (to be very accurate, this method needs a large number of satellites and is, thus, more efficient with multiple constellations). Shadow matching can be very effective to reduce cross-street error (if the LOS/NLOS classification algorithm is reliable), but is ineffective in the case of along-street errors.

2.4. Auto-Correlation Function

The Auto-Correlation Function (ACF) is a bit different from the other GNSS indicators presented so far. While they are all available at the output of the receiver, ACF is not accessible as an output and is only available inside the receiver.

ACF is a convolution between the non-distorted local signal replica generated by the receiver and the potentially distorted signal that comes from the satellite [59]. If the two signals are in-phase, which means that both signals are aligned and that the original one is not distorted, the ACF will have a triangular shape with a perfect symmetry between the positive and negative slope. If the satellite signal is distorted, then this triangular shape will be deformed. A good explanation of how the ACF evolves in the case of LOS and NLOS multipath can be seen in Figure 3. The ACF is based on three correlators (early, prompt, and late), which will slightly change the phase when the replica needs to be correlated with the satellite signal. From the example of Figure 3, we can see different patterns:

- In the case of NLOS (middle of Figure 3):
 - Attenuation of the ACF signal
 - Delay in the maximum of correlation (prompt)
- In the case of multipath (right of Figure 3):
 - Attenuation or augmentation of the ACF amplitude
 - Suppression of the symmetry between the positive and negative slope



Figure 3. Different shapes of the ACF in LOS (left), NLOS (middle), and multipath situations (right).

López-Salcedo et al. [8] proposed to use this slope asymmetry to detect multipath. They begin with the normalization of every function by an estimation of C/N0 in order to be invariant to noise/attenuation and synchronize each sample. Next, the slopes are estimated by the least squares on the three points that are the closest to the maximum of correlation. Then, the values of the two slopes are summed up to create a Slope AsyMmetry indicator (SAM). The SAM value becomes large in urban canyons where multipath occurs, while it stays around zero in the open sky situation. Since the goal of this paper was only to described the GPS signal in harsh environments, we have no information about the success rate of this method. However, an extension of this work showed that the technique was functional for real-time application [60].

Other ACF-based techniques were proposed. Gang et al. [61] used the wavelet transform modulus maxima to detect an "abnormal" ACF shape. The solution was only tested on data coming from the simulation and not tried on real data. Mubarak et al. [14] proposed the ELP (Early Late Phase) indicator, which is the phase difference between the outputs of the early and late correlators. This value becomes high in the presence of multipath and is a good discriminator. This solution was only tested with simulation data. Mubarak et al. [62] showed the compatibility of the method with BOC (Binary Offset Carrier) and BPSK (Binary Phase-Shift Keying) modulation.

Even if detecting multipath by these techniques can be a good insight to get the context, this is not ideal since they are not able to distinguish NLOS from multipath. We can only say that the vehicle is in an environment where GNSS admits some error, but cannot give a precise context. For instance, in the case where there is a majority of NLOS satellite signals, the vehicle is probably in an urban area, and the context is not likely to change soon. On the other hand, if the majority of satellites have the multipath effect (which can be due to some particular materials close to the receiver, buildings, water, and so on), even if the signal quality is not the best, the GNSS can still give a decent position solution. It is then clear that we cannot properly adapt the data fusion if we base the system only on multipath detection.

Skournetou and Lohan [22] introduced an indicator for indoor/outdoor classification called the Level Crossing Rate (LCR), which basically counts the number of times the ACF crosses a threshold (denoted *a*) both from below and above. It is mathematically defined as follows:

$$LCR(a) = \operatorname{card}\left\{i | (\overline{\mathcal{R}}_i \le a \land \overline{\mathcal{R}}_{i+1} > a) \qquad \bigwedge \qquad (\overline{\mathcal{R}}_{i+1} \le a \land \overline{\mathcal{R}}_i > a)\right\}$$
(2)

 \mathcal{R}_i is the non-coherent average ACF at timestamp defined as:

$$\overline{\mathcal{R}}\left(\widehat{\tau},\widehat{f}_{D}\right) = \frac{1}{N_{nc}} \sum_{N_{nc}} \left| \frac{1}{N_{c}} \sum_{m=1}^{N_{c}} \mathcal{R}\left(\widehat{\tau},\widehat{f}_{D},m\right) \right|^{2}$$
(3)

 N_{nc} is the number of non-coherent integrations (meaning that the phase is not used during the correlation step since the correlation signal is squared up), while N_c is the number of coherent integrations (meaning that the phase is used during the correlation step since the original correlation signal is used). \hat{f}_D is the Doppler frequency. $\hat{\tau}$ is the delay, and *m* an epoch. Then, a new indicator called the averaged ACF level of max LCR, which actually corresponds to the averaged ACF level where the LCR is at its highest value, is computed. This indicator will be the threshold for discriminating outdoors from indoors. The threshold value was computed after various simulations with C/N0 around 20 dB.Hz (which corresponds to an intermediate environment between indoors and outdoors). They found a threshold for different values of N_{nc} and N_c . Then, if the value computed for the upcoming signal is lower than the averaged threshold, the receiver is indoors, otherwise it is outdoors. The performance of this detection system varied greatly, between a 67% and 95% successful classification rate depending on the movement of the receiver and the number of multipaths signals. All the tests were conducted in simulation.

2.5. Combination of Multiple Indicators

For now, except the ones in [19,20], all the GNSS-based context detection solutions presented so far were based on a single indicator. This part presents techniques that combine several indicators for context detection. Most of them are based on Machine Learning (ML) approaches.

For LOS/NLOS classification, Yozevitch et al. [49] built a binary decision tree thanks to the RapidMiner software. Their tree was based on C/N0, pseudo-range residuals, and satellite elevations. The learning phase was done by using a labelled database created with an Android application where the user can take a picture and compare the satellite elevation with the building boundaries.

They achieved a success rate of around 80%. Their results were compared to those of an expectation maximization algorithm, and a similar performance was achieved.

Hsu [55] proposed a method to classify the GNSS signals into three categories: NLOS, multipath, and LOS (clean signal). This is interesting because the number of NLOS satellites could help us to extract some context information like indoors, urban, or open sky. The authors, based in Hong Kong, set up an antenna in a dense building location during 24 h. They labelled the recorded GNSS data between NLOS or LOS by applying the ray tracing method using a 3D terrain model. They used those data as the training population for the SVM (Support Vector Machine) classifier with the following four different features:

- Signal strength (C/N0 value)
- Change rate of the received signal strength
- Pseudo-range residuals
- Difference between delta pseudo-range (5) and pseudo-range rate (6), defined as follows (4):

$$|\Delta \rho - \dot{\rho} \cdot \Delta t| \tag{4}$$

where $\Delta \rho$ is the difference between the actual and previous pseudo-range measurements:

$$\Delta \rho_k^{(i)} = \rho_k^{(i)} - \rho_{k-1}^{(i)} \tag{5}$$

and $\dot{\rho}$ is the difference in Doppler shift (f_{Doppler} is the Doppler shift, *c* the celerity, and f_{L1} the GPS L1 band carrier frequency):

$$\dot{\rho}^{(i)} = f_{\text{Doppler}}^{(i)} - \frac{c}{f_{L1}} \tag{6}$$

With a population of 85,365 samples and using an SVM classifier on the signal strength and the difference between delta pseudo-range and pseudo-range rate, they achieved a 75% correct classification. Using all four features did not improve the classification performance, so we can fairly say that the features used here are not the best set.

Xu et al. [63] also tried to classify GNSS signals into three different categories, LOS, NLOS, and multipath, using to an SVM model. They compared three different classifiers. The first one, called NMEA (National Marine Electronics Association), is based on three different GNSS features: C/N0 value, satellites' elevations, and the difference between delta pseudo-range and pseudo-range rate. The second classifier, RINEX (Receiver Independent Exchange Format), is based only on C/N0 and satellites' elevations. The last classifier called "correlator" is based on four features extracted from the ACF: the ratio between the maximum correlation value measured and the standard value (open sky area), the mean and variance of the correlation peak delay, and ELP. The labelling of the training population was done by using a 3D model but with the sky mask and not ray tracing. However, the sky mask does not work very well with satellites whose line-of-sight passes close to the edges of a building and does not detect multipath. The authors proposed to use the double difference of the carrier phase to help the labelling. They trained their RINEX and NMEA model with 6802 samples and their correlator model with 69,000 samples. The final results can be found in Table 3. We can conclude that the classifier on the ACF function works better than the other two. This is quite logical since the ACF gives more information that is far more robust than C/N0 values, which have a large variance. However, we have the same observation as in Section 2.4: accessing the ACF costs much and is not granted for every receiver.

		NMEA-Level Classification Results		RIN Classifi	NEX-Level cation Results	Correlator-Level Classification Results		
		LOS	NLOS	LOS	NLOS	LOS	NLOS	
Labelled	LOS	1194	119498286522		139	1279	17	
Results	NLOS	286			520	179	633	
F1 Score		80.42		78.11		90.39		
Overall Accuracy (%)		81.71		79.67		90.70		

Table 3. Performances of the different algorithms (extracted from [63]). NMEA, National Marine Electronics Association; RINEX, Receiver Independent Exchange Format.

In [9], the authors tried to detect transitions between indoor and outdoor contexts. To do so, four different families were created: deep indoors (no window, no balcony), shallow indoors (the opposite of deep indoors), semi outdoors (outdoors, but many buildings are surrounding the user position), and open outdoors (clear sky). However, the authors of [9] finally decided to only distinguish indoors from outdoors, so the final classification was a simple binary case. In order to classify the signals, they started by selecting 36 different features, which belonged to three categories:

- Spatial geometry
 - Azimuth distribution of the satellite
 - Satellite azimuth distribution proportion
 - Proportion of the number of satellites within a range of 90° of the azimuth
 - Proportion of the number of satellites within a range of 180° of the azimuth
 - Position Dilution of Precision (PDoP), Vertical Dilution of Precision (VDoP), Horizontal Dilution of Precision (HDoP)
- Time sequence
 - The number of visible satellites from time t₂ to time t₁
 - The ratio of satellites, the CNR (Carrier-to-Noise Ratio) of which decreases from time t₂ to time t₁
 - The ratio of satellites, the CNR of which holds from time t₂ to time t₁
 - The ratio of satellites, the CNR of which decreases from time t₂ to time t₁
- Statistical
 - Number of satellites at the current time
 - Mean, variance, standard deviation, minimum, maximum, median, range, interquartile range, skewness, kurtosis of all visible satellites' CNR
 - Mean, variance, standard deviation, minimum, maximum, median, range, interquartile range, skewness, kurtosis of visible satellites' CNR under different sliding window lengths
 - Mean of PDoP, VDoP, and HDoP

Different ML algorithms were applied on those 36 features: RF (Random Forest), SVM, AdaBoost (or Adaptive Boosting), XGBoost, and LightGBM (Gradient Boosting Machine). Another algorithm called "stacking" was also tried. This algorithm combines all the outputs of the previous ML algorithms except SVM to train a new layer called the meta-classifier. Finally, they tried a last classifier based on the stacking one, but with an HMM as the final layer. Four different datasets, which represented more than 195,851 samples, were created for their classification test. Those datasets were acquired at different locations, with different scenarios and sensors. The best classification results were obtained for almost every test by the last classifier, the Stacking algorithm with the HMM model. Using all three types of features increased the success rate of classification, but by a very small margin. Using

12 of 30

only spatial or temporal features led to a similar result. LightGBM was the most efficient single model in terms of classification. The limit of this article was that they only focused on indoor/outdoor classification and not on a more precise environmental context (urban, open-field, etc.).

Overall, the main drawback of supervised ML is that you will need a good labelling. As stated in the different articles, it is very challenging to correctly label the acquired data since it is difficult to have a ground truth for GNSS signals (LOS, NLOS, multipath). To us, it seems clear that unsupervised ML will have better performance, but will require a bigger database, which is also difficult to create. It is also important to mention that during our research, we did not find any open source GNSS database. More globally, the use of ML to classify NLOS, LOS, and multipath appears to be difficult. As for the indoor/outdoor classifier [9] it cost more computational resources than the detector seen in [20] and was less precise (since the other method can detect indoors, outdoors, urban, and open sky).

2.6. Summary of GNSS Indicators

In this section, we have seen that many GNSS indicators that are available as the output of the receiver (C/N0, elevations, pseudo-range) can be useful in terms of context detection, but can be unreliable in certain situations. For instance, a simple threshold on C/N0 can give an idea about the context, but cannot be used as a standalone indicator. Then, we paid attention to custom indicators on the ACF function. Those indicators give much information since the ACF function provides information on the quality of the signal (NLOS, LOS, multipath), which is not affordable with simple indicators like C/N0. However, the ACF function is only available inside the receiver, and that is a huge drawback. We then reviewed the solutions that try to combine multiple indicators to see if they are more reliable than simple indicators and easier to implement than ACF ones. Most of those solutions are based on machine learning. Those techniques show good results and are able to classify LOS/NLOS without requiring access to the ACF function. However we highlighted that it is very difficult to use supervised machine learning since the labelling step is very complicated.

At this time, there is no current GNSS-based solution that can extract every context of interest in a reliable way. Another type of data is needed to perform a robust classification; that is why, in the next section, we will introduce vision-based context indicators.

3. Vision-Based Context Indicators

This section reviews camera-based indicators that can give information on the environmental context. The main idea is to find methods that can help or be complementary to those based on GNSS. The first part focuses on the detection of NLOS satellites thanks to sky extraction. The second part reviews scene analysis methods that can be useful to detect if we are in indoor or outdoor situations. Then, the third and fourth parts present satellite and aerial imagery that can be useful to perform a context cartography. Finally, the last part focuses on the combination of the vision-based indicators.

3.1. Sky Extraction

Sky extraction has been widely investigated for various applications, like meteorology [64,65], navigation [66], or even to help scene analysis [67]. Those techniques are based on different setups including visible, Ultraviolet (UV), and Infrared (IR) wavelengths, which all have their pros and cons. Far-IR reduces illumination variations, while Near IR (NIR) increases contrast between clouds and sky. UV wavelengths seem to be interesting for sky segmentation, but need to be filtered at specific peaks [68]. If sky segmentation has received so much attention, this is due to the difficulty of the task since there are many variations due to weather or time [69]. Indeed, segmenting sky into sunny or cloudy situations is totally different. Furthermore, daylight illuminance will have different characteristics than dawn or dusk illuminance (in terms of luminosity, colour temperature, shadows, etc.). Various methods exist to achieve sky segmentation, but many of them cannot be applied for our navigation purposes due to a computational time that is too high. We can cite the example of the graph cut-based solution [70] or scene parsing [71].

Sky extraction for NLOS detection in GNSS navigation has also received much attention [10–13]. Its main idea is to implement a wide angle sky-oriented camera on top of a vehicle to extract the sky part of the image. Then, satellites that appear outside the sky area when projecting their positions on the image are classified as NLOS. Those satellites are filtered so that the GNSS navigation system only uses LOS satellites in the position calculation. This considerably improves the precision of the GNSS position solution, but in certain contexts like urban canyons, only a couple of satellites are LOS; therefore, the position cannot even be computed. It is important to notice that this technology has been developed to improve the position accuracy and not to detect a context. However, based on this technology, it seems straightforward to extract some context information. As already said in the previous section, having the number of NLOS satellites can give an idea of the vehicle environment. If there are many NLOS satellites, this implies that the drone is in an urban or indoor context. If there is no NLOS satellite, we are basically in an open sky situation. From the sky extraction result, we can also create other indicators, like the presence of sky or not, the sky segment area, or the shapes of the sky boundaries. For instance, when we narrowed down the possible context to either indoors or urban canyon, the sky area indicator can be used to differentiate the two contexts. Several setups are presented below. Some are based on visible wavelengths and others on IR wavelengths. Note that there is no setup available based on UV filtering, as it is very difficult to assemble such a setup on board a vehicle.

Gakne and Petovello [11] used a camera with a standard lens (not a fisheye one) and a sensor working in the visible frequencies. The objective had an FOV (Field Of View) of 90°, which is quite small for a system operating at 360° (satellites can appear everywhere around the user position). They tried different image processing algorithms to extract the sky: Otsu, mean shift, graph cut, HMRF-EM (Hidden Markov Random Fields-Expectation-Maximization). In the end, the simplest algorithm, Otsu, proved to be the best in terms of both the processing time and the segmentation accuracy (cf. Table 4). The authors of [11] concluded that removing NLOS satellites from the position solution calculation was not a good idea since there were too few LOS satellites in many cases. However, this conclusion was due to the limited FOV of their camera and hence cannot be generalized.

	Average Processing	Accuracy (%)			
Algorithm	Time Per Image (s)	Sunny	Cloudy		
Otsu	0.015	80.8	94.7		
Mean shift	35.5	55.4	90.5		
HMRF-EM	73.9	36.3	82.7		
Graph cut	1.8	59.8	82.8		

Table 4. The performances of the different sky extraction algorithms (extracted from [11]).

The authors of [10] used a visible camera coupled with a fisheye lens. The paper also started with a comparison of different sky extraction algorithms: colour and texture mix, mathematical morphology, k-means segmentation. The comparison was done on a database of 100 cloudy and sunny images at various locations: urban area, under trees, inside a tunnel. The mathematical morphology outperformed the other algorithms and was actually the quickest one as well (0.4 s). This algorithm started by removing the contrast parameter on each pixel of the image to pre-segment the latter. The pre-segmented image still had illuminance variations due to the sun and/or clouds. To remove these illuminance variations and obtain a homogeneous area, a geodesic reconstruction by dilatation was used. This sky segmentation algorithm reached a 93% good classification rate with a database only composed of 100 pictures of sky, but with none semi-indoor/indoor pictures where the ceiling could be classified as sky. This method is particularly efficient on high contrast images but struggles in low contrast situations which occur often in outdoor, as shown in [72]. As we not only aim to filter NLOS satellites, but use sky segmentation to help classify the context, sky segmentation must be reliable in

every situation. The satellites are projected onto the image by using the STK GNSS simulation software. (Figure 4).



Figure 4. Original image (**left**) and its sky segmentation representation (**right**) with projected satellites (green = LOS, red = NLOS).

Their final results also showed the fact that there were less than four LOS satellites in many situations. This geodesic method of sky extraction was also used in [73]. Firstly, the NLOS satellites are removed based on the sky extraction results. Secondly, a multipaths mitigation algorithm is applied, and finally, the position solution is calculated thanks to the Vector Delay Frequency Lock Loop (VDFLL).

A complementary work to [10] was given in [74]. In this paper, they built a region classifier in four steps. The first step is to pre-process the input image with a colorimetric invariant transform. Then, the image is segmented into regions, and features are computed (essentially based on histograms) for each region. Finally, they classified regions to extract only the sky ones using a maximal similarity algorithm based on the Bhattacharyya distance. To compare the similarity, references needed to be compute beforehand. That is why two learning databases (sky and background) needed to be created. This method reached a classification rate beyond 99% on the same database as in [10] and outperformed other supervised and unsupervised classifiers (Fisher, KNN, SVM, fuzzy C-means, SRM, etc.).

Meguro et al. [13] used another hardware setup with an omnidirectional infrared camera. This choice was firstly because an infrared camera can segment the sky very easily since it appears without any illumination variations (see Figure 5).



Figure 5. Infrared original image (left), sky segmentation based on Otsu threshold (right).

Infrared cameras also work at night, which is one of the limitations of visible cameras [75]. According to the authors, the omnidirectional camera was chosen since it is easier to design than a fisheye lens. One problem is that the mirrors are visible in the image. Therefore, we have a zone of uncertainty where information is missing. As done in the two other papers [10,11], they projected the satellites onto the image and concluded on the high probability of having less than four LOS satellites. However, the authors suggested two solutions to this issue. The first one is to use multi-constellation systems, using GPS and Galileo, for instance. However, when the article was written, Galileo was not fully available yet. The second solution is to filter the NLOS satellites depending on their pseudo-range residuals and their C/N0 values. Indeed, some of the satellites that are declared NLOS are sometimes

just next to a sky boundary detected on the image and could be actually LOS. A similar filtering technique based on C/N0 values was also used in [40], but with different image processing. They used the Canny edge detector to create regions. Then, satellites positions were projected onto the images. The area with the highest C/N0 satellite was considered as sky, and flood filling could be started. This method does not seem robust in many situation, since it is based on edge detection and implies that the sky area is uniform (no cloud). For the same objective of NLOS confirmation, we can propose another indicator of the distance between a projected satellite and the sky, also by taking into account the uncertainties in the sky extraction, as well as in the position estimation. Wen et al. [76] proposed to weight NLOS and LOS satellites differently instead of a simple filtering of NLOS satellites. The method showed an improvement in terms of positioning, but the authors did not compare their work with other papers.

Obviously, semantic segmentation, thanks to deep learning, can also be used for sky segmentation on fisheye images. Different solutions already exist [77,78] with, respectively, 94% and 88% correct sky pixel classification. Even if using a deep learning solution on fisheye images with a training database obtained from rectilinear images is convenient (it is possible to use the LabelMe database for instance [79]), it has limited performances [80,81]. To overcome such limitations, a new training database associated with ground truth labelling with fisheye images is required. Another solution would be to modify the rectilinear image with a post-processing operation to create a fisheye-like distortion [78,80]. The main advantage of semantic segmentation is that it can be used to detect additional classes like buildings or trees, which can be beneficial for our algorithm selection purposes. There is still a major interrogation on deep learning solutions since there is no monitoring of the FPS (Frame Per Second) capability.

3.2. Scene Analysis/Classification

Now that we have seen the importance of what is above the vehicle (sky extraction and NLOS detection), we will take a look at what it is facing. Most of the autonomous vehicles have a frontal camera for navigation and/or mission purposes. The art of recognizing what contains a scene is called scene analysis. This is interesting for context detection, as it could help to distinguish indoors or outdoors for instance. Such decision making is referred to as scene classification. Filtering indoor scenes from outdoor ones seems like a simplification of scene classification, but in fact, it is not so simple. Indeed, the type of objects can vary greatly in both situations (plants can be found inside for instance), and external parameters such as illumination or weather can greatly influence the rendering of a picture. Different solutions exist, but there are two main families: the methods based on an analysis of the picture as a whole, which are called holistic methods, and approaches based on local descriptors, which are indicators computed on a sub-part of the image. Most of the time, the sub-parts are created by a segmentation algorithm. There are also a few methods that cannot be applied in our case. One such method is based on the labelling of the different objects of a scene to then classify it depending on the found objects (for example, if a desk, a bed, and a chair are detected in the picture, it can be classified as a bedroom). Such solutions are useful when it is necessary to precisely classify scenes to perform a mission task at a specific place. However, in our navigation application, where we only need to distinguish indoors from outdoors, the level is too high [82]. Lastly, methods based on the metadata of the picture also exist [83,84]. Indeed, with the exif (exchangeable image file) format, it is possible to have access to various parameters, which are referred as metadata. There is a wide range of possible parameters, like GPS position, time, camera manufacturer, or even information, in the picture. Those last parameters are the most interesting for image classification. Here is a non-exhaustive list of them: aperture, autofocus distance, exposure time, f-number, use of the flash, and so on. In our case, we do not have access to all the required data (for instance, it is not possible to use flash since there is no flash on a UGV setup), and therefore, we cannot use this technique.

Various local descriptors have existed in scene analysis for many years: histograms/moments (in different colour spaces: RGB [85], LUV[86,87], LST[67,88], Otha [89,90], HSV [91]), wavelets [67,85,88],

Discrete Cosine Transform (DCT) [89], edge detectors (like Canny [35]), edge orientations [86,87], HOG (Histogram of Oriented Gradients), Scale Invariant Feature Transform (SIFT), etc. Most of them were reviewed in [92]. These local indicators are often combined and used as the input to classifiers. Examples of multi-feature systems with classifiers as an output were presented in [23,24,27]. The most frequent classifiers are KNN and SVM. There are also much more complex existing solutions. Some are based on brightness, gamut, exposure, and gain [28], some on a stacking of multiple indicators (in various domains), and deep learning models [29,93,94]. It is also important to note that multiple databases exist to train models (which all have their pros and cons: number of images, low resolution, image size, etc.): IITM-SCID2 [95], the fifteen scene categories [96], SUN [97], INRIA Holidays [98], Antonio Torralba's Indoor [99,100], MIT places [101].

Holistic methods are presented as robust methods to classify scenes since deep learning solutions compare themselves to those. Holistic methods have also been compared to human recognition performances and have achieved very similar results in terms of correct classifications [102]. This result can seem very counter intuitive since the features used in those methods are global properties, which are not based on any colour or textural information, but show the potential of holistic-based technique. Two different holistic methods were presented in [25,26].

The first one is called GIST. This technique is inspired by human vision and how the human brain computes information [103,104]. When a human looks at a scene, he/she will recognize the scene in less than 200 ms. This means that in a very short time period, a human can analyse a large amount of perceptual information. The goal of GIST is to create a representation in which it is easy and quick to find out what contains the scene. GIST is computed using 32 Gabor filters (with eight different orientations and four different scales). From this operation, thirty-two feature maps of the same resolution as the initial image can be obtained. Each image is divided into 16 regions using a 4×4 grid filter. In each of those regions, the gradient is computed (orientation and scale). Then, each of the 16 sub-regions is concatenated to compute the average value. At the end, a 512 (32×16)-dimensional vector is obtained and is our GIST descriptor. A representation of GIST is shown in Figure 6.



Figure 6. Original image and its GIST representation (indoor example on the **top**, outdoor example on the **bottom**).

Tahir et al. [25] used a feed-forward neural network to classify their GIST vectors between indoors and outdoors. A 90.8% successful classification rate was obtained and, according to the authors,

could be further improved by adding new local and global features. They also proposed to test other classifiers such as SVM to see if there was any improvement.

The second holistic method is called CENsus TRansform hISTogram (CENTRIST) in which the authors tried to extract the structural properties of the image and remove the textural information. A first transformation called Census Transfrom (CT) is applied on every pixel of the image. The CT makes a comparison of the value of each of the eight neighbouring pixels with the value of the centre pixel. If the neighbour's value is higher than the central value, then this neighbour pixel is set to zero, otherwise to one. Once this operation is done on the eight neighbours, a binary number is created starting from the top-left pixel to the bottom-right one (see Figure 7). This binary number is then converted to a decimal number. The centre pixel value is set to this value, which is called Census Transform value (CT value). The CT value has the advantage of being free from illumination and gamma variations. The next step is to compute the histogram of the CT values. Then, the classification is performed by the SVM and 1-NN classifiers. The last one had better results and should therefore be exploited. The classification performance of CENTRIST was compared to those of GIST and SIFT, and the result showed that CENTRIST was more reliable and faster (those comparisons were done with different datasets than the one used in the GIST article). In this comparison, the authors showed that GIST works better in outdoor situations than in indoor ones, which is not surprising since GIST was firstly developed to classify outdoor scenes [105]. Quattoni and Torralba [100] also showed the limitation of GIST to classify indoor scenes. However, CENTRIST still has some limitations. It is not rotation invariant, which is not a big problem for us since the camera and the scene are always orientated in the same direction. Besides, the colour information is totally excluded, which is surprising since it could help the scene classification greatly. Meng et al. [31] proposed an extension of CENTRIST, where the local difference magnitude is computed. The results showed a slight improvement in terms of classification.

125	130	180		1	1	0	
125	160	125	\Rightarrow	1		1	$\implies (11011011)_2 \implies (219)_{10}$
220	90	145]	0	1	1	

Figure 7. Illustration of the census transform.

Finally, some solutions try to mix holistic and classical approaches. Ganesan and Balasubramanian [32] used GIST in the Ohta colour space combined with CENTRIST as input features of an SVM classifier and achieved high classification performance on the IITM SCID2 and MIT datasets. Balasubramanian and Anitha [33] mixed SIFT features with enhanced GIST ones and classified the MIT-67 database with an SVM model.

In summary, scene analysis can be a good complement to GNSS signal indicators to classify the indoor from the outdoor context. Most of them are based on deep learning and present good results. Even holistic methods are now used as features for deep learning solutions. However, in the previous articles, there was no indication of the processing time of those solutions, which is one of the main limitation for navigation systems.

3.3. Satellite Imagery

Another way to know in which context the vehicle is evolving is to take some altitude and look at its surroundings from the sky. The goal is to create a context map of the vehicle's environment thanks to online open-access satellite imagery databases. Much work has been done to classify the environment with remote sensing [106–108]. Most of the methods use the same technique, which consists of the segmentation of areas, the creation of the descriptors (shape, colour, and texture) of those areas, and finally, classification using data mining. Other solutions that detect specific attributes like buildings [109], roads [110,111], or trees [112] also exist. Satellite imagery performs well when you want to look at a large field of view at a coarse resolution.

For our objective of vehicle navigation, we need a metric of precision to distinguish such contexts as under a tree, close to a building, etc. Since satellite imagery does not fulfil this requirement, we decided to investigate aerial photography, which has a finer resolution.

3.4. Aerial Photography

Little work has been recently done on aerial photography since it is an alternative to satellite imagery with a much higher resolution and pixel intensities. Aerial photography actually works in a very similar manner to the satellite imagery solutions except that satellite imagery is very often hyperspectral, which is rarely the case for aerial photography. This creates major challenges to achieve a high resolution segmentation for Aerial Photography (AP) since only RGB channels are available and also some objects may have a similar aspect, but are very different semantically. For instance, tree regions are very close to grass or bush regions from a textural and colour point of view, but semantically totally different. The opposite is also true, as for example cars, which have different shapes and colours, but are semantically the same.

In the first example of [15], they tried to detect urban and wild land contexts with two different approaches: object-based and pixel-based approaches. The pixel-based algorithm uses ISODATA, an unsupervised classifier that uses the minimum spectral distance. Then, a 3×3 majority filter is used to find a class. As for the object-based method, the first step is to segment the image. To do so, the authors used the software called Definiens eCognition, which has a bottom-up algorithm based on spectral, textural, and user parameters. The classification is performed using a multi-scale system. The large segmented areas are classified by applying a fuzzy algorithm, while the smaller objects are classified by a KNN (K-Nearest- Neighbours) algorithm. At the end, the pixel-based method had a 62% correct classification rate, while the object-based one had 80%. The authors concluded that an object-based algorithm is more accurate than a pixel-based one for remote sensing.

Like many fields of research, AP classification can be performed using CNN, as done in [17], where AP images were classified into five classes: building, low vegetation, tree, car, impervious surface. To do so, the authors built a network where features (which were obtained by a slightly modified version of the MobileNetV2 network [113]) were shared to extract two different channels: semantic and boundary maps. The results were promising, but the main issue of this approach is that they used the NIR channel, which helps to extract grass and trees. This channel is not available in most AP databases. It has also been shown that using different scale features with fully convolutional network helps to achieve a better detection [114].

Another example of segmentation and classification of aerial images was proposed in [16]. This method is also object-based. A super-pixel algorithm (TurboPixels [115]) is applied on the image to obtain a first segmentation. Then, two features (textural and spectral) are extracted. The textural descriptor used is the LBP-HF (Local Binary Pattern Histogram Fourier) [116]. The spectral feature used is the colour histogram in the RGB domain (tests were done also in HSV, but gave poor results). Finally, two different KNN models, one for the colour and one for the texture, classify the super-pixel into a label: building, road, tree, grass, and water. The models were trained with manual segmentation, and the results were promising. Resolution can be improved with an increase in the number of super-pixels. In the article, the number of super-pixels was set to 500 without any explanation. Perhaps an increase in the super-pixels leads to more computation consumption.

It should be noted that a context map based on aerial photography can be pre-loaded on a vehicle since in most of the situations, its operation zone is known a priori. With a such solution, we will be able to have knowledge about the surrounding context if the vehicle position is estimated precisely enough with respect to the context map resolution.

3.5. Combination of Vision-Based Techniques

To our knowledge, there is no article in the literature that proposes to combine different vision-based techniques to extract context. This is not surprising since the sky extraction methods

presented earlier are not used for the purposes of context detection, but only for NLOS detection in GNSS navigation. Furthermore, satellite imagery is used in cartography more than for context detection.

3.6. Summary of Vision-Based Indicators

We have seen in this section different vision-based indicators for environmental context extraction. The first part focused on different methods of sky segmentation. Methods using a colour camera and basic image processing algorithms show good results in outdoor scenarios. Semantic segmentation solutions based on deep learning also perform well. Nevertheless, there is no fisheye camera database for sky detection available, so it has to be created manually or from a rectilinear image with an additional transformation. The methods based on IR facilitate the image processing since the sky is already segmented, but the IR camera does not have the large FOV that the visible camera + fisheye lens setup has. Indeed, it is very difficult to implement such optics on an IR sensor. Although it is possible to use an omnidirectional camera as an alternative, it creates a blind spot since mirrors appear on the image. The main use of sky segmentation is satellite mitigation, by projecting a satellite's position on the segmented image to identify LOS or NLOS. Filtering NLOS satellites improves the accuracy of the navigation solution, but also leads to the impossibility of computing a position when less than four satellites remain. Many solutions exist to solve this problem: using multiple constellations, improving NLOS detection using an association of image processing and GNSS indicators, admitting uncertainty about the projected satellite position, and so on. Sky segmentation can also be used as additional information for outdoor context detection.

Next, we focused on visual scene analysis, which can help in the outdoor/indoor classification. There are three different types of methods: local descriptors+classifier, holistic approaches+classifier, and deep learning. Local descriptors have been used for many years, but showed a lack of robustness. Hence, they were replaced by holistic methods, which consider the image as a whole to reduce variability. Currently, state-of-the-art methods combine holistic descriptors with a deep learning network. Although this achieves a high rate of correct classification, we have no information on the computation time of those techniques and if they are suitable for navigation application.

In the third and fourth part, we reviewed satellite and aerial imagery semantic segmentation with the goal to build a map of contexts. It appeared that satellite imagery resolution is too low for our navigation purposes, and hence, we focused on aerial imagery. Two different methods exist: object-based and pixel-based. The object-based algorithms have the advantage of classifying area by area, but have the drawback of requiring a pre-processing step to find each area of interest. The pixel-based techniques do not need any pre-processing, but the spatial structure object is lost, and it creates a salt and pepper effect. CNN can help to resolve this matter, but most of the proposed deep learning solutions used a database with the NIR channel, which helps to segment trees and vegetation. Therefore, we cannot confirm the classification rates of those methods.

To conclude on vision-based context indicators, the majority of works dealt with place recognition or image segmentation, but few of them were applied to the context-aware navigation.

4. Context Detection Based on Other Sensors

The two previous sections covered GNSS- and vision-based indicators that can be used for context detection. We are still missing indicators on the IMU, our last sensor available in the classical navigation setup. Actually, the IMU does not give interesting information on the context, as it functions in the same manner regardless of the environment. Using the IMU measurement enables detecting pedestrian steps [117], road vibrations (to check if the car is moving) [118], or how straight a vehicle is going thanks to its gyro [119]. Measurements from accelerometers and gyroscopes are also used in more complex systems to identify the behavioural context and the activity of smartphone users (for example, running, climbing stairs, inside a lift [120]), to create an activity map for example [121]. Such information will not directly help our environmental context detection. Nonetheless, using IMU

data is still the best and simplest way to know if the vehicle is in motion or not. This information could be interesting for managing context detection because if the drone is idle, we have no reason to look for a new context (it is however still important to follow the evolution of the satellite constellation).

It has been shown in past years that a depth map could help the indoor/outdoor classification [30]. Generally, the depth map should be computed from a stereo vision setup, which is far from being implemented on every ground vehicle. In the previous article, the depth map has been estimated (thanks to random Markov fields) due to the lack of a stereoscopic dataset. The depth map can also be extracted thanks to LiDAR or RGB-D cameras. Papers based on those sensors are presented in the following paragraphs.

Considering LiDAR (LIght Detection And Ranging) or an RGB-D camera, it has been shown that such a sensor could be used to perform indoor/outdoor classification. The LiDAR sensor is often used on drones and other autonomous vehicles to create Digital Elevation Models (DEMs). It is usually used in addition to the standard visible camera since features cannot be found in non-textured environments. For example, in [122], they tried to detect windows using camera-LiDAR fusion. Such detection results could give us information on how far the vehicle is from windows or potentially from the exit/entrance of the buildings. Börcs et al. [123] trained a CNN model to classify buildings, vehicles, pedestrians, ground, and road clutter from a 3D point cloud, which could help us to detect outdoor scenes. Lim and Suter [18] combined features from LiDAR and a camera (heights, colours, spin image, estimated normals) to label super-voxels of a 3D point cloud as five classes (trees, trunk, building, pathway, grass) using a multi-scale conditional random field. This method could be useful for the context mapping task, but for online navigation use, it has a drawback compared to the AP method since it cannot be pre-loaded. However, it is important to note that LiDAR has not been used as a stand alone sensor for context detection so far.

RGB-D cameras, which provide colour and depth images, are also used for context detection, but only in indoor applications. Most of the time, they are used in scenes where there is a wide range of possible objects and where pixel data are not sufficient to label them. Adding another type of information with the depth map helps to label objects. Such a method is referred to as scene labelling (see [124]). Once objects have been labelled, a classifier is used to find the context depending on the objects in the scene. For instance, Gupta et al. [125] classified rooms thanks to objects detection by using an SVM classifier on the histogram of oriented gradients of the depth map. RGB-D cameras are not very useful in our case of vehicle navigation, since this technology is not reliable in outdoor environments. If any depth map data are needed for context detection, then it would be preferable to use LiDAR.

The following methods are used on smartphone devices where multiple sensors are embedded and hence are not really applicable to classical navigation systems without modifications.

Xu et al. [126] combined GNSS features (number of visible satellites, positioning error value) and light sensor data to build a decision tree with multiple thresholds. The system classified data into three families: indoor, outdoor, semi-outdoor. Both static and dynamic tests were conducted, and it appeared that in dynamic environments, the solution admitted a higher error.

References [1,127,128] proposed methods based on WiFi signals where multiple features are computed (signal-to-noise ratio, number of accessible emission points). In [1], WiFi signal measurements were combined with a GNSS-based indicator. Such multi-sensor-based context detection was also proposed in [129], where WiFi was combined this time with Bluetooth. Even more complex solutions exist such as [7,130]. Ali et al. [7] used light, internal clock, received signal strength indicator cell tower, and WiFi features in different independent modules to classify each epoch as indoors, urban outdoors, rural outdoors, and underground. Zhou et al. [130] used proximity, magnetic field, acceleration, and WiFi sensor information to detect indoor and outdoor contexts using an HMM.

Wang et al. [131] used Global System for Mobile (GSM) features to classify four different contexts: open outdoors, semi-outdoors, light indoors, and deep indoors. The authors computed the mean, standard deviation, maximum, minimum, and range for every received signals. Then, they tried

machine learning algorithms to classify their data and concluded that the random forest method was the one achieving the highest score in every performance indicator (accuracy, sensitivity, precision). The big advantages of using GSM are that the signal is available basically everywhere and has very low consumption in terms of energy.

Audio can also be used to classify indoors from outdoors, as shown in [132]. The idea is to emit a sound during 100ms with variable frequencies and then capture its reverberation. Once the recording is done, Mel-Frequency Cepstral Coefficients (MFCC) are calculated and used as inputs to an SVM classifier, which is trained with labelled data. The system reached 96.26% correct classification.

Barometers can be used to detect context. For instance, they are used to determine if the setup is in motion, walking, or idle [133]. Barometers are also used for indoor navigation [134] by detecting different floors. The main advantage of the barometer is its very low consumption, but it has also drawbacks, which reduce its performances like weather/temperature changes or vibrations.

A magnetic sensor is also useful for indoor/outdoor classification. Indeed, Ashraf et al. [135] used a naive Bayes classifier with magnetic intensity features (mean, variance, kurtosis, median, standard deviation, interquartile, percentiles, squared deviation, trimmed mean, coefficient of variance, average absolute deviation) as inputs to define if the system was located inside or outside. The system achieved a classification accuracy of 83.26%.

Finally, Krumm and Hariharan [136] tried to classify indoors from outdoors by using a temperature sensor. Three different statistical distributions were computed for indoors, outdoors, and ambient temperature. Based on those distributions with ambient temperature measurements, the probabilities of being inside or outside were computed. The system had a success rate of 81%.

In this section, we reviewed different types of sensors that could be used for the environmental context detection. The diversity of those techniques shows that context detection cannot be defined precisely and that the process to find the correct classification is not deterministic. Basically, every sensor can be used since it will always add more information. It is then important to select the sensors that are more useful and reliable for the targeted application. For our navigation purposes, this paper focuses on the classical GNSS + IMU + Vision sensor setup while keeping in mind that the above listed sensors can be useful when another source of information is required.

5. Summary of the Different Solutions and Perspectives

This paper overviewed existing environmental context detection solutions mainly based on GNSS and vision sensors, from a navigation point of view. We firstly focused on the GNSS-based indicators and saw that C/N0 and the K-Rician factor are good indicators to detect indoor/outdoor contexts. The negative part is that a certain amount of integration time is needed to compute those values. It is also important to note that those indicators are highly sensitive and not robust to environmental variations (loss of satellites, sudden attenuation, etc.). We then took a look at the pseudo-range residual and its second derivative, which suffer from the same problems as C/N0: sensitive to noise and a need much integration time. Nonetheless, these values can be useful to distinguish LOS from NLOS satellites. Knowing the number of NLOS satellites would give information on the context: the more NLOS satellites there are, the higher the chance of being indoors (or in urban canyons) will be. Another indicator to confirm the LOS/NLOS classification (which is also important information for shadow matching) is the satellite elevation. Indeed, if the satellite is low in the sky, there is a higher chance for the signal to be blocked (NLOS). We then introduced the ACF indicators. This function is much more complicated to extract since it is only accessible inside the GNSS receiver and not as a simple output like the previous indicators. Different solutions exist, but most of them only detect multipaths without differentiating this from NLOS, which is not sufficient for context detection purposes. Finally, we presented different solutions based on multivariate GNSS data. Most of those solutions apply machine learning techniques and seem to suffer from the difficulty in creating a labelled database. Furthermore, often, the classification is just among indoors, outdoors, and intermediate, which is not fine enough for navigation adaptation purposes.

In the second part of this paper, we showed the different methods to extract context using vision information. The first method is to extract the sky segment (on an image of a wide angle camera) and to project the satellites' positions on it in order to classify LOS/NLOS satellites. This provides us with two major pieces of information: the number of NLOS satellites and the presence of sky. Such information can give an idea about the context (indoors, urban, open sky, etc.). The sky extraction enables the GNSS navigation to improve its positioning accuracy by detecting and excluding the NLOS satellites. However, this step must be done with caution since it can lead to too few satellites available, which makes the GNSS position estimation impossible. We have seen multiple indicators that can be used in addition to help the satellite filtering like C/N0 or pseudo-range. The second vision-based solution exposed is to use a frontal camera and to analyse the scene in order to classify it and detect context. Different methods exist from local descriptors to the holistic method. Both techniques will at the end need a classifier. The third approach is to use satellite imagery, or aerial photography. We quickly saw that Satellite Imagery (SI) is not precise enough in terms of resolution for our vehicle navigation purposes, and thus, AP should be used. Most of the segmentation algorithms for AP are similar to those for SI. Those AP-based techniques are interesting since they can help us detect basically every type of context, including trees and water, that the GNSS indicators cannot (the only way to detect trees or water with GNSS is to use reflectometry, which is not possible in our case, since the receiver is near the ground level and in motion). It is also important to mention that object-based image analysis is more powerful than pixel-based (which creates a salt and pepper effect since there is no region segmentation) in the case of remote sensing, as proven in [137].

A summary table of all the reviewed methods of context detection is given in Table 5. This table is only based on the literature and not on any tests, which can explain our doubts about certain indicators (denoted as "?").

Indicators/Context		Open Sky	Low Density Urban	Dense Urban	Urban Canyon	Light Indoor	Deep Indoor	Light Forest	Dense forest	Water
	C/N0	~	~	~	~	~	~	?	?	×
GNSS	K-Rician	~	~	~	~	~	~	?	?	×
	Pseudo-range	?	?	?	?	?	?	?	?	×
	Satellite elevation	?	?	?	?	?	?	?	?	×
Vision	Sky extraction	~	?	?	?	~	~	?	?	×
	No. of NLOS satellites	~	~	~	~	~	~	~	~	×
	Aerial photography	~	?	?	?	?	?	~	~	~
	Scene classification	~	×	×	×	~	~	×	×	×

(✔ = useful; ¥ = not useful; ? = unclear)

6. Conclusions and Future Work

We showcased in this paper that current environmental context detection methods are not robust since using a single dimension descriptor is not sufficient. One of the perspectives towards robust context detection is to fuse different indicators from multiple sensors and to create a multi-dimensional context detector. To our knowledge, so far, there is no such solution in the literature trying to combine vision and GNSS indicators. This is essentially due to the fact that most of the existing works on context detection are for applications implemented on smartphones and that the use of cameras on those devices is complicated (it is not possible to have a sky-orientated camera on a smartphone for instance).

Moreover, a fair comparison between all the presented methods in this review is not possible at this time as there is no publicly available environmental context database for GNSS or visual modalities. Indeed, each work cannot be reproduced as each used its own very small dataset that was not provided to the community.

After this literature review on the context detection methods, our future work includes the development of a new method to detect environmental context consistently based primarily on vision, but with the aid of GNSS indicators for navigation adaptation purpose. As a function of the detected context, we can select the best sensor set to use or adapt to fuse measurements (for example, GNSS loose-/tight-couplings) in the navigation filter, as well as choose a more efficient algorithm to extract the information in the vision system (as shown in Figure 1). Such a context-aware adaptive navigation filter will enhance its localization accuracy and availability in a complex environment. Our new context detection method will also be applicable to make a context mapping for offline use in mission/path planning tasks.

Author Contributions: Conceptualization, F.F., D.V. and Y.W.; methodology, F.F., D.V. and Y.W.; software, F.F.; validation, D.V. and Y.W.; formal analysis, F.F., D.V. and Y.W.; investigation, F.F.; resources, F.F., D.V. and Y.W.; data curation, F.F.; writing original draft preparation, F.F.; writing review and editing, F.F., Y.W. and D.V.; visualization, D.V. and Y.W.; supervision, D.V. and Y.W.; project administration, D.V. and Y.W.; funding acquisition, D.V. and Y.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Defense Innovation Agency (AID) of the French Ministry of Defense, Research Project CONCORDE N° 2019 65 0090004707501.

Acknowledgments: The testing part of this work have been conducted using materials from both Isae-Supaero and Onera. Thanks for the feedback of the different reviewers and the help of MDPI concerning the paper formatting.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Groves, P.; Martin, H.; Voutsis, K.; Walter, D.; Wang, L. Context Detection, Categorization and Connectivity for Advanced Adaptive Integrated Navigation. In Proceedings of the 26th International Technical Meeting of the Satellite Division of the Institute of Navigation, ION GNSS 2013, Nashville, TN, USA, 16–20 September 2013; Volume 2, pp. 15–19.
- Groves, P.D.; Wang, L.; Walter, D.; Martin, H.; Voutsis, K.; Jiang, Z. The four key challenges of advanced multisensor navigation and positioning. In Proceedings of the 2014 IEEE/ION Position, Location and Navigation Symposium—PLANS 2014, Monterey, CA, USA, 5–8 May 2014; pp. 773–792. [CrossRef]
- 3. Gao, H.; Groves, P.D. Improving environment detection by behavior association for context-adaptive navigation. *Navigation* **2020**, *67*, 43–60. [CrossRef]
- Frank, K.; Nadales, M.J.V.; Robertson, P.; Angermann, M. Reliable Real-Time Recognition of motion related human activities using MEMS inertial sensors. In Proceedings of the 23rd International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS 2010), Portland, OR, USA, 21–24 September 2010.
- Pei, L.; Chen, R.; Liu, J.; Kuusniemi, H.; Chen, Y.; Tenhunen, T. Using Motion-Awareness for the 3D Indoor Personal Navigation on a Smartphone. In Proceedings of the 24th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS 2011), Portland, OR, USA, 20–23 September 2011; Volume 4.
- 6. Esmaeili Kelishomi, A.; Garmabaki, A.; Bahaghighat, M.; Dong, J. Mobile User Indoor-Outdoor Detection Through Physical Daily Activities. *Sensors* **2019**, *19*, 511. [CrossRef] [PubMed]
- Ali, M.; ElBatt, T.; Youssef, M. SenseIO: Realistic Ubiquitous Indoor Outdoor Detection System Using Smartphones. *IEEE Sens. J.* 2018, 18, 3684–3693. [CrossRef]
- López-Salcedo, J.A.; Parro-Jimenez, J.; Seco-Granados, G. Multipath detection metrics and attenuation analysis using a GPS snapshot receiver in harsh environments. In Proceedings of the 2009 3rd European Conference on Antennas and Propagation, Berlin, Germany, 23–27 March 2009; pp. 3692–3696.
- 9. Zhu, Y.; Luo, H.; Wang, Q.; Zhao, F.; Ning, B.; Ke, Q.; Zhang, C. A Fast Indoor/Outdoor Transition Detection Algorithm Based on Machine Learning. *Sensors* **2019**, *19*, 786. [CrossRef]
- Attia, D.; Meurie, C.; Ruichek, Y.; Marais, J.; Flancquart, A. Image analysis based real time detection of satellites reception state. In Proceedings of the 13th International IEEE Conference on Intelligent Transportation Systems, Funchal, Madeira Island, Portugal, 19–22 September 2010; pp. 1651–1656. [CrossRef]

- Gakne, P.; Petovello, M. Assessing image segmentation algorithms for sky identification in GNSS. In Proceedings of the 2015 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Banff, AB, Canada, 13–16 October 2015; pp. 1–7. [CrossRef]
- 12. Marais, J.; Meurie, C.; Attia, D.; Ruichek, Y.; Flancquart, A. Toward accurate localization in guided transport: Combining GNSS data and imaging information. *Transp. Res. Part C Emerg. Technol.* **2013**, *43*. [CrossRef]
- 13. Meguro, J.; Murata, T.; Takiguchi, J.; Amano, Y.; Hashizume, T. GPS Multipath Mitigation for Urban Area Using Omnidirectional Infrared Camera. *IEEE Trans. Intell. Transp. Syst.* **2009**, *10*, 22–30. [CrossRef]
- Mubarak, O. Analysis of early late phase for multipath mitigation. In Proceedings of the 21st International Technical Meeting of the Satellite Division of the US Institute of Navigation, Savannah, GA, USA, 10–12 December 2008; pp. 16–19.
- Cleve, C.; Kelly, M.; Kearns, F.R.; Moritz, M. Classification of the wildland–urban interface: A comparison of pixel- and object-based classifications using high-resolution aerial photography. *Comput. Environ. Urban Syst.* 2008, 32, 317–326. [CrossRef]
- Ghiasi, M.; Amirfattahi, R. Fast semantic segmentation of aerial images based on color and texture. In Proceedings of the Iranian Conference on Machine Vision and Image Processing, MVIP, Zanjan, Iran, 10–12 September 2013; pp. 324–327. [CrossRef]
- 17. Zhang, G.; Lei, T.; Cui, Y.; Jiang, P. A Dual-Path and Lightweight Convolutional Neural Network for High-Resolution Aerial Image Segmentation. *ISPRS Int. J. Geo-Inf.* **2019**, *8*, 582. [CrossRef]
- 18. Lim, E.H.; Suter, D. 3D terrestrial LIDAR classifications with super-voxels and multi-scale Conditional Random Fields. *Comput.-Aided Des.* **2009**, *41*, 701–710. [CrossRef]
- Lin, T.; O'Driscoll, C.; Lachapelle, G. Development of a Context-Aware Vector-Based High-Sensitivity GNSS Software Receiver. *Proc. Int. Tech. Meet. Inst. Navig.* 2011, 2, 1043–1055.
- 20. Gao, H.; Groves, P. Environmental Context Detection for Adaptive Navigation using GNSS Measurements from a Smartphone. *Navigation* **2018**, *65*, 99–116. [CrossRef]
- 21. Chen, K.; Tan, G. SatProbe: Low-energy and fast indoor/outdoor detection based on raw GPS processing. In Proceedings of the IEEE INFOCOM 2017—IEEE Conference on Computer Communications, Atlanta, GA, USA, 1–4 May 2017; pp. 1–9. [CrossRef]
- Skournetou, D.; Lohan, E.S. Indoor location awareness based on the non-coherent correlation function for GNSS signals. In Proceedings of the Finnish Signal Processing Symposium, FINSIG'07, Oulu, Finland, 30 August 2007.
- 23. Deng, J.; Zhang, J. Combining Multiple Precision-Boosted Classifiers for Indoor-Outdoor Scene Classification. *Inform. Technol. Appl.* **2005**, *1*, 720–725. [CrossRef]
- 24. Raja, R.; Roomi, S.M.M.; Dharmalakshmi, D.; Rohini, S. Classification of indoor/outdoor scene. In Proceedings of the 2013 IEEE International Conference on Computational Intelligence and Computing Research, Enathi, India, 26–28 December 2013; pp. 1–4. [CrossRef]
- Tahir, W.; Majeed, A.; Rehman, T. Indoor/Outdoor Image Classification Using GIST Image Features and Neural Network Classifiers. In Proceedings of the 2015 12th International Conference on High-capacity Optical Networks and Enabling/Emerging Technologies (HONET), Islamabad, Iran, 21–23 December 2015. [CrossRef]
- Wu, J.; Rehg, J.M. CENTRIST: A Visual Descriptor for Scene Categorization. *IEEE Trans. Pattern Anal. Mach. Intell.* 2011, 33, 1489–1501. [CrossRef] [PubMed]
- Raja, R.; Roomi, S.M.M.; Dharmalakshmi, D. Robust indoor/outdoor scene classification. In Proceedings of the 2015 Eighth International Conference on Advances in Pattern Recognition (ICAPR), Kolkata, India, 4–7 January 2015; pp. 1–5.
- Lipowezky, U.; Vol, I. Indoor-outdoor detector for mobile phone cameras using gentle boosting. In Proceedings of the 2010 IEEE Computer Society Conference on Computer Vision and Pattern Recognition—Workshops, CVPRW 2010, San Francisco, CA, USA, 13–18 June 2010; pp. 31–38. [CrossRef]
- Chen, C.; Ren, Y.; Kuo, C.C. Large-Scale Indoor/Outdoor Image Classification via Expert Decision Fusion (EDF). In Proceedings of the Computer Vision—ACCV 2014 Workshops, Singapore, 1–5 November 2014; pp. 426–442. [CrossRef]
- Pillai, I.; Satta, R.; Fumera, G.; Roli, F. Exploiting Depth Information for Indoor-Outdoor Scene Classification. In Proceedings of the Image Analysis and Processing—ICIAP 2011, Ravenna, Italy, 14–16 September 2011; Volume 2, pp. 130–139. [CrossRef]

- 31. Meng, X.; Wang, Z.; Wu, L. Building global image features for scene recognition. *Pattern Recognit.* **2012**, 45, 373–380. [CrossRef]
- Ganesan, A.; Balasubramanian, A. Indoor versus outdoor scene recognition for navigation of a micro aerial vehicle using spatial color gist wavelet descriptors. *Vis. Comput. Ind. Biomed. Art* 2019, 2, 20. [CrossRef] [PubMed]
- 33. Balasubramanian, A.; Anitha, G. Indoor Scene Recognition for Micro Aerial Vehicles Navigation using Enhanced-GIST Descriptors. *Def. Sci. J.* **2018**, *68*, 129–137. [CrossRef]
- Seco-Granados, G.; López-Salcedo, J.A.; Jimenez-Banos, D.; Lopez-Risueno, G. Challenges in Indoor Global Navigation Satellite Systems: Unveiling its core features in signal processing. *IEEE Signal Process. Mag.* 2012, 29, 108–131. [CrossRef]
- 35. Payne, A.; Singh, S. Indoor vs. outdoor scene classification in digital photographs. *Pattern Recognit.* 2005, 38, 1533–1545. [CrossRef]
- 36. Ma, C.; Jee, G.I.; MacGougan, G.; Lachapelle, G.; Bloebaum, S.; Cox, G.; Garin, L.; Shewfelt, J. Gps signal degradation modeling. In Proceedings of the International Technical Meeting of the Satellite Division of the Institute of Navigation, Salt Lake City, UT, USA, 11–14 September 2001.
- Lehner, A.; Steingaß, A. On land mobile satellite navigation performance degraded by multipath reception. In Proceedings of the 2007 European Navigation Conference (ENC-GNSS), ENC-GNSS 2007, Geneva, Switzerland, 29–31 May 2007.
- MacGougan, G.; Lachapelle, G.; Klukas, R.; Siu, K. Degraded GPS Signal Measurements with A Stand-Alone High Sensitivity Receiver. In Proceedings of the 2002 National Technical Meeting of the Institute of Navigation, San Diego, CA, USA, 28–30 September 2002.
- Gómez-Casco, D.; López-Salcedo, J.A.; Seco-Granados, G. Optimal Post-Detection Integration Technique for the Reacquisition of Weak GNSS Signals. *IEEE Trans. Aerosp. Electron. Syst.* 2019, 56, 2302–2311. [CrossRef]
- Matera, E.R.; Garcia Peña, A.J.; Julien, O.S.; Milner, C.; Ekambi, B. Characterization of Line-of-Sight and Non-Line-of-Sight Pseudorange Multipath Errors in Urban Environment for GPS and Galileo. In *ITM* 2019, *International Technical Meeting of The Institute of Navigation*; ION: Reston, VA, USA, 2019; pp. 177–196. [CrossRef]
- 41. Tranquilla, J.M.; Carr, J.P. GPS Multipath Field Observations at Land and Water Sites. *Navigation* **1990**, 37, 393–414. [CrossRef]
- 42. Fenton, P.; Jones, J. The Theory and Performance of NovAtel Inc.'s Vision Correlator. In Proceedings of the 18th International Technical Meeting of the Satellite Division of The Institute of Navigation, Long Beach, CA, USA, 13–16 September 2005.
- Bhuiyan, M.Z.H.; Lohan, E.S.; Renfors, M. Code Tracking Algorithms for Mitigating Multipath Effects in Fading Channels for Satellite-Based Positioning. *EURASIP J. Adv. Signal Process.* 2008, 2008, 863629. [CrossRef]
- 44. Sahmoudi, M.; Amin, M.G. Fast Iterative Maximum-Likelihood Algorithm (FIMLA) for Multipath Mitigation in the Next Generation of GNSS Receivers. *IEEE Trans. Wirel. Commun.* **2008**, *7*, 4362–4374. [CrossRef]
- 45. Spangenberg, M.; Julien, O.; Calmettes, V.; Duchâteau, G. Urban Navigation System for Automotive Applications Using HSGPS, Inertial and Wheel Speed Sensors. OATAO, 2008. Available online: https://oatao.univ-toulouse.fr/3183/ (accessed on 23 April 2008).
- 46. Vincent, F.; Vilà-Valls, J.; Besson, O.; Medina, D.; Chaumette, E. Doppler-aided positioning in GNSS receivers—A performance analysis. *Signal Process.* **2020**, *176*, 107713. [CrossRef]
- Azemi, G.; Senadji, B.; Boashash, B. Estimating the Ricean K-factor for mobile communication applications. In Proceedings of the Seventh International Symposium on Signal Processing and Its Applications, Paris, France, 1–4 July 2003; Volume 2, pp. 311–314.
- 48. Kumar, G.; Rao, G.; Kumar, M. GPS Signal Short-Term Propagation Characteristics Modeling in Urban Areas for Precise Navigation Applications. *Positioning* **2013**, *4*, 192–199. [CrossRef]
- Yozevitch, R.; Moshe, B.B.; Weissman, A. A Robust GNSS LOS/NLOS Signal Classifier. *Navigation* 2016, 63, 429–442. [CrossRef]
- Piñana-Diaz, C.; Toledo-Moreo, R.; Bétaille, D.; Gómez-Skarmeta, A.F. GPS multipath detection and exclusion with elevation-enhanced maps. In Proceedings of the 2011 14th International IEEE Conference on Intelligent Transportation Systems (ITSC), Washington, DC, USA, 5–7 October 2011; pp. 19–24.

- 51. Miura, S.; Hisaka, S.; Kamijo, S. GPS multipath detection and rectification using 3D maps. In Proceedings of the 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), The Hague, The Netherlands, 6–9 October 2013; pp. 1528–1534. [CrossRef]
- 52. Groves, P.; Jiang, Z.; Wang, L.; Ziebart, M. Intelligent Urban Positioning using Multi-Constellation GNSS with 3D Mapping and NLOS Signal Detection. In Proceedings of the 25th International Technical Meeting of the Satellite Division of the Institute of Navigation (Ion Gnss 2012), Nashville, TN, USA, 17–21 September 2012; Volume 1, pp. 458–472.
- Peyraud, S.; Bétaille, D.; Renault, S.; Ortiz, M.; Mougel, F.; Meizel, D.; Peyret, F. About Non-Line-Of-Sight Satellite Detection and Exclusion in a 3D Map-Aided Localization Algorithm. *Sensors* 2013, 13, 829–847. [CrossRef]
- 54. Groves, P.; Jiang, Z. Height Aiding, C/N0 Weighting and Consistency Checking for GNSS NLOS and Multipath Mitigation in Urban Areas. *J. Navig.* **2013**, *66*, 653–669. [CrossRef]
- Hsu, L. GNSS multipath detection using a machine learning approach. In Proceedings of the 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC), Yokohama, Japan, 16–19 October 2017; pp. 1–6. [CrossRef]
- 56. Groves, P. Shadow Matching: A New GNSS Positioning Technique for Urban Canyons. J. Navig. 2011, 64, 417–430. [CrossRef]
- 57. Wang, L.; Groves, P.D.; Ziebart, M. GNSS Shadow Matching Using a 3D Model of London in Urban Canyons. In Proceedings of the European Navigation Conference 2011, London, UK, 28 November–1 December 2011.
- 58. Yozevitch, R.; Moshe, B.b. A Robust Shadow Matching Algorithm for GNSS Positioning. *Navigation* **2015**, 62, 95–109. [CrossRef]
- Pagot, J.B.; Thevenon, P.; Julien, O.; Gregoire, Y.; Amarillo-Fernandez, F.; Maillard, D. Estimation of GNSS Signals Nominal Distortions from Correlation and Chip Domain. In Proceedings of the 2015 International Technical Meeting of The Institute of Navigation, Dana Point, CA, USA, 26–28 January 2015.
- 60. Egea, D.; Seco-Granados, G.; López-Salcedo, J.A. Comprehensive Overview of Quickest Detection Theory and its Application to GNSS Threat Detection. *Giroskopiya Navig.* **2016**, *95*, 76–97. [CrossRef]
- 61. Li, G.; Li, G.x.; Lv, J.; Chang, J.; Jie, X. A method of multipath detection in navigation receiver. In Proceedings of the 2009 International Conference on Wireless Communications Signal Processing, Nanjing, China, 13–15 November 2009; pp. 1–3. [CrossRef]
- Mubarak, O. Performance comparison of multipath detection using early late phase in BPSK and BOC modulated signals. In Proceedings of the 2013 7th International Conference on Signal Processing and Communication Systems, ICSPCS 2013—Proceedings, Carrara, VIC, Australia, 16–18 December 2013. [CrossRef]
- 63. Xu, B.; Jia, H.; Luo, Y.; Hsu, L.T. Intelligent GPS L1 LOS/Multipath/NLOS Classifiers Based on Correlator-, RINEX- and NMEA-Level Measurements. *Remote Sens.* **2019**, *11*, 1851. [CrossRef]
- 64. Klebe, D.; Blatherwick, R.; Morris, V. Ground-based all-sky mid-infrared and visible imagery for purposes of characterizing cloud properties. *Atmos. Meas. Tech.* **2014**, *7*, 637. [CrossRef]
- 65. Dev, S.; Lee, Y.H.; Winkler, S. Color-Based Segmentation of Sky/Cloud Images From Ground-Based Cameras. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* **2017**, *10*, 231–242. [CrossRef]
- Meguro, J.; Murata, T.; Amano, Y.; Hashizume, T.; Takiguchi, J.i. Development of a Positioning Technique for an Urban Area Using Omnidirectional Infrared Camera and Aerial Survey Data. *Adv. Robot.* 2008, 22, 731–747. [CrossRef]
- 67. Serrano, N.; Savakis, A.E.; Luo, J. Improved scene classification using efficient low-level features and semantic cues. *Pattern Recognit.* **2004**, *37*, 1773–1784. [CrossRef]
- Stone, T.; Mangan, M.; Ardin, P.; Webb, B. Sky segmentation with ultraviolet images can be used for navigation. In Proceedings of the Conference: Robotics: Science and Systems 2014, Berkeley, CA, USA, 12–16 July 2014. [CrossRef]
- 69. Mihail, R.P.; Workman, S.; Bessinger, Z.; Jacobs, N. Sky segmentation in the wild: An empirical study. In Proceedings of the 2016 IEEE Winter Conference on Applications of Computer Vision (WACV), Lake Placid, NY, USA, 7–10 March 2016; pp. 1–6.
- 70. Tao, L.; Yuan, L.; Sun, J. SkyFinder: Attribute-based sky image search. *ACM Trans. Graph.* **2009**, *28*, 68. [CrossRef]

- Liu, C.; Yuen, J.; Torralba, A. Nonparametric scene parsing: Label transfer via dense scene alignment. In Proceedings of the 2009 IEEE Conference on Computer Vision and Pattern Recognition, Miami, FL, USA, 20–25 June 2009; pp. 1972–1979.
- 72. Nafornita, C.; David, C.; Isar, A. Preliminary results on sky segmentation. In Proceedings of the International Symposium on Signals, Circuits and Systems (ISSCS), Iasi, Romania, 9–10 July 2015.
- 73. Shytermeja, E.; Pena, A.G.; Julien, O. Proposed architecture for integrity monitoring of a GNSS/MEMS system with a Fisheye camera in urban environment. In Proceedings of the International Conference on Localization and GNSS 2014 (ICL-GNSS 2014), Helsinki, Finland, 24–26 June 2014; pp. 1–6.
- 74. El Merabet, Y.; Ruichek, Y.; Ghaffarian, S.; Samir, Z.; Boujiha, T.; Touahni, R.; Messoussi, R. Horizon Line Detection from Fisheye Images Using Color Local Image Region Descriptors and Bhattacharyya Coefficient-Based Distance. In Proceedings of the International Conference on Advanced Concepts for Intelligent Vision Systems, Lecce, Italy, 24–27 October 2016; Volume 10016. [CrossRef]
- 75. Chapman, L.; Thornes, J.E.; Muller, J.; McMuldroch, S. Potential Applications of Thermal Fisheye Imagery in Urban Environments. *IEEE Geosci. Remote Sens. Lett.* **2007**, *4*, 56–59. [CrossRef]
- 76. Wen, W.; Bai, X.; Kan, Y.C.; Hsu, L. Tightly Coupled GNSS/INS Integration via Factor Graph and Aided by Fish-Eye Camera. *IEEE Trans. Veh. Technol.* **2019**, *68*, 10651–10662. [CrossRef]
- 77. Tsai, Y.H.; Shen, X.; Lin, Z.; Sunkavalli, K.; Yang, M.H. Sky is not the limit: Semantic-aware sky replacement. *ACM Trans. Graph.* **2016**, *35*, 1–11. [CrossRef]
- Deng, L.; Yang, M.; Qian, Y.; Wang, C.; Wang, B. CNN based semantic segmentation for urban traffic scenes using fisheye camera. In Proceedings of the 2017 IEEE Intelligent Vehicles Symposium (IV), Los Angeles, CA, USA, 11–14 June 2017; pp. 231–236.
- 79. Russell, B.; Torralba, A.; Murphy, K.; Freeman, W. LabelMe: A Database and Web-Based Tool for Image Annotation. *Int. J. Comput. Vis.* **2008**, *77*, 157–173. [CrossRef]
- 80. Blott, G.; Takami, M.; Heipke, C. Semantic Segmentation of Fisheye Images. In Proceedings of the European Conference on Computer Vision (ECCV) Workshops, Munich, Germany, 8–14 September 2018.
- Böker, C.; Niemeijer, J.; Wojke, N.; Meurie, C.; Cocheril, Y. A System for Image-Based Non-Line-Of-Sight Detection Using Convolutional Neural Networks. In Proceedings of the IEEE Intelligent Transportation Systems Conference (ITSC), Auckland, New Zealand, 27–30 October 2019; pp. 535–540
- Khan, S.H.; Hayat, M.; Bennamoun, M.; Togneri, R.; Sohel, F.A. A Discriminative Representation of Convolutional Features for Indoor Scene Recognition. *IEEE Trans. Image Process.* 2016, 25, 3372–3383. [CrossRef] [PubMed]
- Boutell, M.; Luo, J. Beyond pixels: Exploiting camera metadata for photo classification. *Pattern Recognit*. 2005, *38*, 935–946. [CrossRef]
- Boutell, M.; Luo, J. Bayesian fusion of camera metadata cues in semantic scene classification. In Proceedings of the 2004 IEEE Computer Society Conference on Computer Vision and Pattern Recognition, CVPR 2004, Washington, DC, USA, 27 June–2 July 2004; Volume 2, pp. 1063–6919.
- 85. Gupta, L.; Pathangay, V.; Patra, A.; Dyana, A.; Das, S. Indoor versus Outdoor Scene Classification Using Probabilistic Neural Network. *EURASIP J. Adv. Signal Process.* **2007**, 2007, 1–10. [CrossRef]
- Tao, L.; Kim, Y.H.; Kim, Y.T. An efficient neural network based indoor-outdoor scene classification algorithm. In Proceedings of the 2010 Digest of Technical Papers International Conference on Consumer Electronics (ICCE), Las Vegas, NV, USA, 9–13 January 2010; pp. 317–318.
- Vailaya, A.; Figueiredo, M.; Jain, A.; Zhang, H.J. Content-based hierarchical classification of vacation images. In Proceedings of the IEEE International Conference on Multimedia Computing and Systems, Florence, Italy, 7–11 June 1999; Volume 1, pp. 518–523.
- Serrano, N.; Savakis, A.; Luo, A. A computationally efficient approach to indoor/outdoor scene classification. In Proceedings of the Object Recognition Supported by User Interaction for Service Robots, Quebec City, QC, Canada, 11–15 August 2002; Volume 4, pp. 146–149.
- Szummer, M.; Picard, R.W. Indoor-outdoor image classification. In Proceedings of the 1998 IEEE International Workshop on Content-Based Access of Image and Video Database, Bombay, India, 3 January 1998; pp. 42–51.
- Luo, J.; Savakis, A. Indoor vs outdoor classification of consumer photographs using low-level and semantic features. In Proceedings of the 2001 International Conference on Image Processing, Thessaloniki, Greece, 7–10 October 2001; pp. 745–748. [CrossRef]

- 91. Kim, W.; Park, J.; Kim, C. A Novel Method for Efficient Indoor–Outdoor Image Classification. *Signal Process. Syst.* 2010, *61*, 251–258. [CrossRef]
- 92. Tong, Z.; Shi, D.; Yan, B.; Wei, J. A Review of Indoor-Outdoor Scene Classification. In Proceedings of the 2017 2nd International Conference on Control, Automation and Artificial Intelligence (CAAI 2017), Sanya, China, 25–26 June 2017. [CrossRef]
- Wu, R.; Wang, B.; Wang, W.; Yu, Y. Harvesting Discriminative Meta Objects with Deep CNN Features for Scene Classification. In Proceedings of the IEEE International Conference on Computer Vision (ICCV), Santiago, Chile, 7–13 December 2015.
- Zhou, B.; Lapedriza, A.; Xiao, J.; Torralba, A.; Oliva, A. Learning Deep Features for Scene Recognition using Places Database. In Proceedings of the Advances in Neural Information Processing Systems, Montreal, QC, Canada, 8–13 December 2014; pp. 487–495.
- 95. Extended Scene Classification Image Database Version 2 (IITM-SCID2). Available online: http://www.cse. iitm.ac.in/~vplab/SCID/ (accessed on 4 July 2020).
- 96. The Fifteen Scene Categories. Available online: https://figshare.com/articles/15-Scene_Image_Dataset/ 7007177 (accessed on 4 July 2020).
- 97. SUN Database. Available online: https://groups.csail.mit.edu/vision/SUN/ (accessed on 4 July 2020).
- 98. INRIA Holidays Dataset. Available online: http://lear.inrialpes.fr/people/jegou/data.php (accessed on 4 July 2020).
- 99. Torralba's Indoor Database. Available online: http://web.mit.edu/torralba/www/indoor.html (accessed on 4 July 2020).
- Quattoni, A.; Torralba, A. Recognizing indoor scenes. In Proceedings of the 2009 IEEE Conference on Computer Vision and Pattern Recognition, Miami, FL, USA, 20–25 June 2009; pp. 413–420.
- 101. MIT Places Database. Available online: http://places.csail.mit.edu/index.html (accessed on 4 July 2020).
- 102. Greene, M.R.; Oliva, A. Recognition of natural scenes from global properties: Seeing the forest without representing the trees. *Cogn. Psychol.* **2009**, *58*, 137–176. [CrossRef] [PubMed]
- 103. Oliva, A.; Torralba, A. Building the gist of a scene: The role of global image features in recognition. *Prog. Brain Res.* **2006**, 155, 23–36.
- 104. Oliva, A. Gist of the Scene. In *Neurobiology of Attention*; Elsevier: Amsterdam, The Netherlands, 2005; Volume 696, pp. 251–256. [CrossRef]
- 105. Oliva, A.; Torralba, A. Modeling the Shape of the Scene: A Holistic Representation of the Spatial Envelope. *Int. J. Comput. Vis.* **2001**, *42*, 145–175. [CrossRef]
- 106. Hu, Q.; Wenbin, W.; Xia, T.; Yu, Q.; Yang, P.; Li, Z.; Song, Q. Exploring the Use of Google Earth Imagery and Object-Based Methods in Land Use/Cover Mapping. Remote Sens. 2013, 5, 6026–6042. [CrossRef]
- 107. Kabir, S.; He, D.C.; Sanusi, M.A.; Wan Hussina, W.M.A. Texture analysis of IKONOS satellite imagery for urban land use and land cover classification. *Imaging Sci. J.* **2010**, *58*, 163–170. [CrossRef]
- Tabib Mahmoudi, F.; Samadzadegan, F.; Reinartz, P. Object Recognition based on the Context Aware Decision Level Fusion in Multi Views Imagery. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* 2014, 8, 12–22. [CrossRef]
- Ok, A.O.; Senaras, C.; Yuksel, B. Automated Detection of Arbitrarily Shaped Buildings in Complex Environments From Monocular VHR Optical Satellite Imagery. *IEEE Trans. Geosci. Remote. Sens.* 2013, 51, 1701–1717. [CrossRef]
- Maurya, R.; Gupta, P.; Shukla, A. Road extraction using K-Means clustering and morphological operations. In Proceedings of the 2011 IEEE International Conference on Image Information Processing (ICIIP), Brussels, Belgium, 11–14 September 2011; pp. 1–6. [CrossRef]
- 111. Maboudi, M.; Amini, J.; Hahn, M.; Saati, M. Road Network Extraction from VHR Satellite Images Using Context Aware Object Feature Integration and Tensor Voting. *Remote Sens.* **2016**, *8*, 637. [CrossRef]
- 112. Ardila, J.; Bijker, W.; Tolpekin, V.; Stein, A. Context–sensitive extraction of tree crown objects in urban areas using VHR satellite images. *Int. J. Appl. Earth Obs. Geoinf.* **2012**, *15*, 57–69. [CrossRef]
- Sandler, M.; Howard, A.; Zhu, M.; Zhmoginov, A.; Chen, L.C. MobileNetV2: Inverted Residuals and Linear Bottlenecks. arXiv 2018, arXiv:1801.04381.
- 114. Hu, S.; Ning, Q.; Chen, B.; Lei, Y.; Zhou, X.; Yan, H.; Zhao, C.; Tang, T.; Hu, R. Segmentation of Aerial Image with Multi-scale Feature and Attention Model. In *Artificial Intelligence in China*; Springer: Berlin/Heidelberg, Germany, 2020; pp. 58–66. [CrossRef]

- Levinshtein, A.; Stere, A.; Kutulakos, K.N.; Fleet, D.J.; Dickinson, S.J.; Siddiqi, K. TurboPixels: Fast Superpixels Using Geometric Flows. *IEEE Trans. Pattern Anal. Mach. Intell.* 2009, *31*, 2290–2297. [CrossRef]
- 116. Ahonen, T.; Matas, J.; He, C.; Pietikäinen, M. Rotation Invariant Image Description with Local Binary Pattern Histogram Fourier Features. In *Image Analysis*; Salberg, A.B., Hardeberg, J.Y., Jenssen, R., Eds.; Springer: Berlin/Heidelberg, Germany, 2009; pp. 61–70.
- 117. Frank, K.; Vera, M.J.; Robertson, P.; Angermann, M. Reliable Real-Time Recognition of Motion Related Human Activities Using MEMS Inertial Sensors. In Proceedings of the ION GNSS 2010, Portland, OR, USA, 24–24 September 2010; Volume 4.
- 118. Ramanandan, A.; Chen, A.; Farrell, J.; Suvarna, S. Detection of stationarity in an inertial navigation system. In Proceedings of the 23rd International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS 2010), Portland, Oregon, 1–24 September 2010; Volume 1, pp. 238–244.
- 119. Martí, E.; Martín Gómez, D.; Garcia, J.; de la Escalera, A.; Molina, J.; Armingol, J. Context-Aided Sensor Fusion for Enhanced Urban Navigation. *Sensors* **2012**, *12*, 16802–16837. [CrossRef]
- 120. Saeedi, S.; Moussa, A.; El-Sheimy, N. Context-Aware Personal Navigation Using Embedded Sensor Fusion in Smartphones. *Sensors* 2014, 14, 5742–5767. [CrossRef]
- 121. Gusenbauer, D.; Isert, C.; Krösche, J. Self-contained indoor positioning on off-the-shelf mobile devices. In Proceedings of the 2010 International Conference on Indoor Positioning and Indoor Navigation, Zurich, Switzerland, 15–17 September 2010, pp. 1–9.
- 122. Zhang, R.; Zakhor, A. Automatic identification of window regions on indoor point clouds using LiDAR and cameras. In Proceedings of the IEEE Winter Conference on Applications of Computer Vision, Steamboat Springs, CO, USA, 24–26 March 2014; pp. 107–114. [CrossRef]
- 123. Börcs, A.; Nagy, B.; Benedek, C. Instant Object Detection in Lidar Point Clouds. *IEEE Geosci. Remote Sens. Lett.* 2017, 14, 992–996. [CrossRef]
- 124. Wang, A.; Lu, J.; Cai, J.; Wang, G.; Cham, T. Unsupervised Joint Feature Learning and Encoding for RGB-D Scene Labeling. *IEEE Trans. Image Process.* **2015**, *24*, 4459–4473. [CrossRef]
- 125. Gupta, S.; Arbelaez, P.; Girshick, R.; Malik, J. Indoor Scene Understanding with RGB-D Images: Bottom-up Segmentation, Object Detection and Semantic Segmentation. *Int. J. Comput. Vis.* 2014, 112, 133–149. [CrossRef]
- 126. Xu, W.; Chen, R.; Chu, T.; Kuang, L.; Yang, Y.; Li, X.; Liu, J.; Chen, Y. A context detection approach using GPS module and emerging sensors in smartphone platform. In Proceedings of the 2014 Ubiquitous Positioning Indoor Navigation and Location Based Service (UPINLBS), Corpus Christ, TX, USA, 20–21 November 2014; pp. 156–163.
- 127. Li, S.; Qin, Z.; Song, H.; Yang, X.; Zhang, R. A lightweight and aggregated system for indoor/outdoor detection using smart devices. *Future Gener. Comput. Syst.* 2017. [CrossRef]
- 128. Shtar, G.; Shapira, B.; Rokach, L. Clustering Wi-Fi fingerprints for indoor–outdoor detection. *Wirel. Netw.* **2019**, *25*, 1341–1359. [CrossRef]
- Zou, H.; Jiang, H.; Luo, Y.; Zhu, J.; Lu, X.; Xie, L. BlueDetect: An iBeacon-Enabled Scheme for Accurate and Energy-Efficient Indoor-Outdoor Detection and Seamless Location-Based Service. *Sensors* 2016, 16, 268. [CrossRef] [PubMed]
- Zhou, P.; Zheng, Y.; Li, Z.; Shen, G. IODetector: A Generic Service for Indoor/Outdoor Detection. ACM Trans. Sens. Netw. 2012, 11, 361–362. [CrossRef]
- 131. Wang, W.; Chang, Q.; Li, Q.; Shi, Z.; Chen, W. Indoor-Outdoor Detection Using a Smart Phone Sensor. *Sensors* **2016**, *16*, 1563. [CrossRef]
- Wang, L.; Roth, J.; Riedel, T.; Beigl, M.; Yao, J. AudioIO: Indoor Outdoor Detection on Smartphones via Active Sound Probing. In Proceedings of the EAI International Conference on IoT in Urban Space, Guimarães, Portugal, 21–22 November 2018; pp. 81–95. [CrossRef]
- 133. Sankaran, K.; Zhu, M.; Guo, X.F.; Ananda, A.L.; Chan, M.C.; Peh, L.S. Using mobile phone barometer for low-power transportation context detection. In Proceedings of the 12th ACM Conference on Embedded Network Sensor Systems, Memphis, TN, USA, 3–6 November 2014; pp. 191–205.
- 134. Vanini, S.; Giordano, S. Adaptive context-agnostic floor transition detection on smart mobile devices. In Proceedings of the 2013 IEEE International Conference on Pervasive Computing and Communications Workshops (PERCOM Workshops), San Diego, CA, USA, 18–22 March 2013; pp. 2–7.

- 30 of 30
- 135. Ashraf, I.; Hur, S.; Park, Y. MagIO: Magnetic Field Strength Based Indoor- Outdoor Detection with a Commercial Smartphone. *Micromachines* **2018**, *9*, 534. [CrossRef]
- 136. Krumm, J.; Hariharan, R. TempIO: Inside/Outside Classification with Temperature. In Proceedings of the Second International Workshop on Man-Machine Symbiotic Systems, Kyoto, Japan, 23–24 November 2004.
- 137. Chen, G.; Hay, G.J.; Carvalho, L.M.T.; Wulder, M.A. Object-based change detection. *Int. J. Remote Sens.* 2012, 33, 4434–4457. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).